315' Copyright Ron Burback 1997More Details on the RPC CallsRon BurbackLecture 18

316' Copyright Ron Burback 1997rpc_server_register_ifnregister the interface to the local dce runtimenARG1: interface structure from IDL—UUID, Version, EPVnARG2: object uuidnARG3: the object�s EPV—can be called many times to establish a family of EPVs, object polymorphismnARG4: error status code

317' Copyright Ron Burback 1997rpc_server_use_all_protseqsnARG1: The size of the network queuenARG2: error status

318' Copyright Ron Burback 1997rpc_server_inq_bindingsnARG1: Binding Vector—The binding vector contains the IP and Interface information. nARG2: Error status

319' Copyright Ron Burback 1997rpc_ns_binding_exportnARG1: Syntax of the entry namenARG2: The CDS name of the servicenARG3: The interfacenARG4: The binding vectornARG5: The object identifiernARG6: Error status.

320' Copyright Ron Burback 1997rpc_ep_registernARG1: The interfacenARG2: The binding vectornARG3: Number of ObjectsnARG4: Vector of UUID for each objectnARG5: Error status

321' Copyright Ron Burback 1997rpc_server_listenARG1: Number of threadsnARG2: Error status